

Author's response to reviews

Title: Dental anomalies: prevalence and associations between them in a large sample of non-orthodontic subjects. A cross-sectional study

Authors:

Giuseppina Laganà (giuseppinalagana@libero.it)

Nicolò Venza (nicolo.venza@yahoo.com)

Ali Borzabadi-Farahani (faraortho@yahoo.com)

Francesco Fabi (frfabi76@gmail.com)

Carlotta Danesi (carli.danesi@gmail.com)

Paola Cozza (paolacozza@alice.it)

Version: 2 Date: 31 Jan 2017

Author's response to reviews:

The manuscript was modified, as you suggested:

- The quality of the written English was improved by a native English-speaking colleague (Dr. Ali Borzabadi-Farahani), the corrections are red written.
- All participants' parents signed a written consent in order to use the anonymous data for scientific publication purpose. In order to guarantee anonymity for all participants, we removed birth date and OP taken date from the data file in the repository link. If it is necessary, we can remove the file at the repository link and we can state that the data has been provided as an additional file.
- The first bullet point in the conclusion was deleted.
- The values in lower left parts of Table 3 were deleted in order to make it easier to understand.